UNIVERSITY^{OF} BIRMINGHAM

Research at Birmingham

Establishing research priorities relating to the longterm impact of TIA and minor stroke through stakeholder-centred consensus

Turner, Grace; Palmer, Ruth; McMullan, Christel; Mathers, Jonathan; Marshall, Tom; Calvert, Melanie

DOI: 10.1186/s40900-018-0089-z

License: Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Turner, GM, Backman, R, McMullan, C, Mathers, J, Marshall, T & Calvert, M 2018, 'Establishing research priorities relating to the long-term impact of TIA and minor stroke through stakeholder-centred consensus', Research Involvement & Engagement, vol. 4, no. 2. https://doi.org/10.1186/s40900-018-0089-z

Link to publication on Research at Birmingham portal

Publisher Rights Statement: Checked for eligibility: 16/02/2018

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

• Users may freely distribute the URL that is used to identify this publication.

• Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?)
Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

Open Access



Establishing research priorities relating to the long-term impact of TIA and minor stroke through stakeholder-centred consensus

Grace M. Turner^{1,2*}, Ruth Backman¹, Christel McMullan¹, Jonathan Mathers¹, Tom Marshall^{1,2} and Melanie Calvert^{1,2}

Plain English summary

What is the problem and why is this important?

Mini-strokes are similar to full strokes, but symptoms last less than 24 h. Many people (up to 70%) have long-term problems after a mini-stroke, such as anxiety; depression; problems with brain functioning (like memory loss); and fatigue (feeling tired). However, the current healthcare pathway only focuses on preventing another stroke and care for other long-term problems is not routinely given. Without proper treatment, people with long-term problems after a mini-stroke could have worse quality of life and may find it difficult to return to work and their social activities.

What is the aim of the research?

We wanted to understand the research priorities of patients, health care professionals and key stakeholders relating to the long-term impact of mini-stroke.

How did we address the problem?

We invited patients, clinicians, researchers and other stakeholders to attend a meeting. At the meeting people discussed the issues relating to the long-term impact of mini-stroke and came to an agreement on their research priorities. There were three stages: (1) people wrote down their individual research suggestions; (2) in smaller groups people came to an agreement on what their top research questions were; and (3) the whole group agreed final research priorities.

What did we find?

Eleven people attended who were representatives for patients, GPs, stroke consultants, stroke nurses, psychologists, the Stroke Association (charity) and stroke researchers, The group agreed on eleven research questions which they felt were the most important to improve health and well-being for people who have had a mini-stroke.

The eleven research questions encompass a range of categories, including: understanding the existing care patients receive (according to diagnosis and geographical location); exploring what optimal care post-TIA/ minor stroke should comprise (identifying and treating impairments, information giving and support groups) and how that care should be delivered (clinical setting and follow-up pathway); impact on family members; and education/training for health care professionals.

(Continued on next page)

* Correspondence: G.Turner.1@bham.ac.uk

²Centre for Patient Reported Outcomes Research, University of Birmingham, Edgbaston, Birmingham B15 2TT, England



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

¹Institute of Applied Health Research, University of Birmingham, Edgbaston, Birmingham B15 2TT, England

(Continued from previous page)

Abstract

Background Clinical management after transient ischaemic attack (TIA) and minor stroke focuses on stroke prevention. However, evidence demonstrates that many patients experience ongoing residual impairments. Residual impairments post-TIA and minor stroke may affect patients' quality of life and return to work or social activities. Research priorities of patients, health care professionals and key stakeholders relating to the long-term impact of TIA and minor stroke are unknown.

Methods Our objective was to establish the top shared research priorities relating to the long-term impact of TIA and minor stroke through stakeholder-centred consensus. A one-day priority setting consensus meeting took place with representatives from different stakeholder groups in October 2016 (Birmingham, UK). Nominal group technique was used to establish research priorities. This involved three stages: (i) gathering research priorities from individual stakeholders; (ii) interim prioritisation in three subgroups; and (iii) final priority setting.

Results The priority setting consensus meeting was attended by 11 stakeholders. The individual stakeholders identified 34 different research priorities. During the interim prioritisation exercise, the three subgroups generated 24 unique research priorities which were discussed as a whole group. Following the final consensus discussion, 11 shared research priorities were unanimously agreed.

The 11 research questions encompass a range of categories, including: understanding the existing care patients receive (according to diagnosis and geographical location); exploring what optimal care post-TIA/minor stroke should comprise (identifying and treating impairments, information giving and support groups) and how that care should be delivered (clinical setting and follow-up pathway); impact on family members; and education/training for health care professionals.

Conclusions Eleven different research priorities were established through stakeholder-centred consensus. These research questions could usefully inform the research agenda and policy decisions for TIA and minor stroke. Inclusion of stakeholders in setting research priorities is important to increase the relevance of research and reduce research waste.

Keywords: Transient ischemic attack, Minor stroke, Research priorities, Stakeholder engagement, Long term effects, United Kingdom

Background

Transient ischaemic attack (TIA) occurs when there is a transient disruption of blood supply to the brain, caused by a temporary blood clot [1]. Symptoms are similar to stroke but short lasting, usually < 1 h [1]. In contrast, minor stroke refers to an episode where obstruction of the blood supply to the brain causes tissue death in the brain, but symptoms are non-disabling and people experience short-term functional recovery [2]. Both TIA and minor stroke patients functionally recover well and arguably these diagnoses are a continuum of each other.

TIAs and minor strokes are common. Over 510,000 people in the United Kingdom (UK) have experienced a TIA or minor stroke [3] and the incidence is rising [4]. Clinical management post-TIA and minor stroke focuses on stroke prevention; however, evidence demonstrates that many patients experience ongoing residual impairments [5–7]. A retrospective analysis of UK primary care records found that TIA patients had increased risk of general practitioners (GPs) consultations for fatigue, anxiety, depression and cognitive impairment, compared to matched controls [7]. Furthermore, two recent systematic reviews have found

relatively high prevalence of these impairments post-TIA and minor stroke [5, 8].

Despite good functional recovery, 'hidden' impairments post-TIA and minor stroke may affect patients' quality of life (QoL) [9–14]. Evidence also suggests that TIA and minor stroke are associated with non-return to work, [9, 10, 15–17] decreased life satisfaction and reduced participation in meaningful activities, such as basic self-care, recreational activities and driving [13, 15, 16, 18]. Furthermore, reduced QoL post-TIA and minor stroke has been found to be associated with stroke recurrence [19].

In 2014 UKs leading stroke charity, the Stroke Association, called for "further research into the long-term effects of TIA" [6]. However, few studies that have evaluated interventions to treat/ manage impairments or improve QoL post-TIA/minor stroke. Our review of the literate identified nine relevant studies which comprised interventions of: education and aerobic exercise (n = 6); [20–25] education (n = 1); [26] exercise (n = 1); [27] and multimodal (telephone, internet and paper) support (n = 1) [28]. Over half of these interventions [20, 23–25, 28] did not primarily aim to improve residual impairments or QoL; primary outcomes were vascular risk factors and unplanned use of the healthcare system. Furthermore, only one study (a protocol) is based in the UK [26] (Canada n = 3, Netherlands n = 1, [27] Norway n = 1, [21] Switzerland n = 1, [22] New Zealand n = 1, [20] Ireland n = 1 [25]). Therefore, there is a need to address the long-term effects of TIA/ minor stroke in a UK setting.

Research priorities of patients, health care professionals and key stakeholders are unknown. Involvement of these stakeholders in shaping the research agenda is important to ensure that research is relevant and applicable to the users of research and, therefore, more likely to be implemented in practice [29]. It is also important to promote partnership with these stakeholders and encourage engagement with research [29]. Furthermore, Chalmers et al. [30] suggests that there is often a mismatch between the priorities of researchers and the priorities of the end users of researchers. They recommend that patients and health care professionals are consulted during the setting of research agendas to increase value and reduce research waste [30]. Our objective was to establish the top shared research priorities relating to the long-term impact of TIA and minor stroke through stakeholder-centred consensus.

Methods

A one-day priority setting consensus meeting took place with representatives from different stakeholder groups in October 2016 (Birmingham, UK). The aim of the event was to establish the top shared research priorities relating to the long-term impact of TIA and minor stroke. A one-day event has the advantage of efficient us of time and resources and reduced burden on attendees.

A number of strategies were used to recruit stakeholders, including: adverts on patient websites (including People in Research, Different Strokes, University of the Third Age); presentations at community events (such as the Birmingham and Black Country Stroke Research Awareness Day) email invites sent to relevant stakeholders identified through internet searches and known contacts; and snowballing. Stakeholders included people with a lived experience of TIA or minor stroke, healthcare professionals, stroke charities and stroke researchers. The aim was to establish the top, shared research priorities relating to the long-term impact of TIA and minor stroke. 'Long-term impact' was defined as post-diagnosis of TIA or minor stroke with no restrictions on length of time after the event.

Nominal group technique was used to establish research priorities. This involved three stages: (i) gathering research priorities from individual stakeholders; (ii) interim prioritisation; and (iii) final priority setting.

Gathering research priorities

Prior to the meeting, individual stakeholders were invited to list unanswered questions about life after TIA or minor stroke using free text questionnaires. This generated a long list of research priorities, only duplicated research priorities were removed.

Interim prioritisation

Interim prioritisation of research questions was conducted in in three subgroups comprising 3-4 stakeholders. The groups discussed their individual research priorities and generated a list of shared group priorities which they then ranked in order of priority. Three researchers (GT, CM and RB) facilitated the subgroup discussion.

Final priority setting

Following interim prioritisation, the sub-groups presented their ranked research questions and there was a while group discussion to compare and debate outcomes of the three groups. The lead researcher (GT) facilitated the discussion. As a whole group, a final consensus on of the shared priority areas for research was agreed.

Data collection and ethical approval

Discussions were audio-recorded using an encrypted digital recorder and nominated scribes documented the research priorities. Ethical consent was gained from the University of Birmingham's Science, Technology, Engineering and Mathematics Ethical Review Committee (reference: ERN_16-1057). Written consent was obtained from all stakeholders prior to the meeting.

Results

The priority setting consensus meeting was attended by 11 stakeholders: three TIA/minor stroke patients; three stroke nurses; one stroke consultant; one general practitioner (GP); one psychologist; one representative from the Stroke Association (charity); and one stroke researcher (Table 1).

The individual stakeholders identified 34 different research priorities. These research priorities were discuss in three subgroups which comprised a mix of representatives from different stakeholder groups (Group 1: patient representative, stroke consultant, psychologist; Group 2: patient representative, researcher and two stroke nurses; Group 3: patient representative). During the interim prioritisation exercise, the three subgroups generated 24 unique research priorities which were discussed as a whole group. Following the final consensus discussion, 11 shared research priorities were unanimously agreed (Table 2).

The majority of research priorities (6/11) related to the management of residual impairments, including: effective follow-up pathways; identifying and treating impairments;

Table 1 Stakeholder characteristics (n = 11)

		Numbe
Age	< 25	0
	25-34	1
	35-44	3
	45-54	2
	55-64	3
	65-74	1
	75-84	1
	≥84	0
Sex	Male	6
Stakeholder category	Healthcare professional	5
	Experienced a TIA	3
	Researcher	1
	Psychologist	1
	Stroke Association representative	1

the role of support groups; and educating healthcare professionals. Similarly, two priorities were associated with provision of information and advice to patients.

There were key discussions about lack of follow-up care for TIA and minor stroke patients and inconsistencies in diagnosis and treatment. There was a clear demand for a standardised follow-up pathway; however, lack of resources and organisational constraints were perceived as barriers. Lack of knowledge and guidance on what advice

Table 2 Top research priorities relating to the long term impact of TIA and minor stroke: consensus from TIA/minor stroke patients and key stakeholders

Top 11 research priorities relating to the long term impact of TIA and minor stroke

- What is the most effective follow-up pathway for TIA/ minor stroke patients?
- What is the best way to identify which TIA/ minor stroke patients will experience ongoing impairments?
- What are the best ways to train or educate healthcare professionals to recognise and understand impairments post-TIA/minor stroke?
- What are the best ways to manage and treat psychological impairment, cognitive impairment and fatigue after TIA/minor stroke?
- Would support groups improve outcomes for people after TIA/ minor stroke?
- What is the best setting to support patients with impairments after TIA/minor stroke (such as TIA clinic, general practice, community setting)?
- What advice should healthcare professions give to TIA/minor stroke patients on return to work and activities?
- What information do people want to receive after they have experienced a TIA/minor stroke?
- What impact does having a TIA/minor stroke have on family members, including children?
- What impact does a diagnosis of 'TIA' vs 'minor stroke' have on patients and their follow-up care?
- How do services which treat or manage impairments post-TIA/ minor stroke differ in different regions of England?

healthcare professionals should provide to patients about return to work and usual activities and the best strategies to identify and manage patients with impairments were identified as barriers to optimal care and, therefore, important research priorities.

Discussion

Stakeholders were consulted to identify and prioritise unanswered research questions relating to the long-term impact of TIA and minor stroke. Eleven different research priorities were established through stakeholder-centred consensus. These research questions should be used to inform the research agenda and policy decisions for TIA and minor stroke.

The 11 research questions encompass a range of categories, including: understanding the existing care patients receive (according to diagnosis and geographical location); exploring what optimal care post-TIA/minor stroke should comprise (identifying and treating impairments, information giving and support groups) and how that care should be delivered (clinical setting and follow-up pathway); impact on family members; and education/ training for health care professionals.

The top agreed research questions provide valuable insight into the priorities of key stakeholders to improve healthcare and quality of life after TIA and minor stroke. Importantly, the priority setting process highlighted that current care is inadequate and there are inconsistencies in the care patients receive and access to care even at a local level.

Researchers should use these research priorities to guide future research proposals which should be codesigned with patients and stakeholders. Furthermore, research priorities should be used by funders to help inform funding decisions. The research questions have implications for policy makers, particularly regarding implementing interventions and follow-up pathways. Therefore, it is essential for researchers to engage with and disseminate to policy and decision makers at local and national levels. In addition to addressing the top research priority questions, future research should also consider potential barriers for implementing change to healthcare post-TIA/minor stroke, particularly capacity and resource constraints.

The main strength of this research priority setting process is that there was good representation from different stakeholder groups, the majority which were patients or healthcare professionals who are the end users of research. Through allowing stakeholders to list and rank priorities individually, views of all stakeholders were represented with limited social acceptability bias. The three interim subgroups each included a patient representative and each subgroup was facilitated by a researcher to ensure that everyone had an equal voice. However, an important limitation is the relatively small sample size of stakeholders, which was restricted due to time and resource constraints for recruitment. Another limitation is that the majority of the stakeholders (9/11) were from the West Midlands; therefore, research priorities may not be generalizable to other regions. A larger, regional priority setting exercise, such as the James Lind Alliance priority setting partnership may generate more generalizable findings. However, the main advantage of a one-day priority setting consensus meeting is time and cost efficiency. The James Lind Alliance approach on average costs >£40,000 and takes between 12 to 18 months, after which research priorities may be outdated. For researchers to realistically be able to identify and act on shared research priorities, it is important to have a robust but time and resource efficient methodology.

From a research team perspective, the priority setting consensus meeting was beneficial, not only to guide the teams future research but also to develop collaborations and networks with different stakeholder groups. We recommend that other research groups adopt a similar approach to identifying research questions for other research areas. The main challenge with organising this event was recruiting stakeholders; we found that, although the event was planned with plenty of notice, work commitments and other unforeseen circumstances meant that a large number of people were unable to attend at the last minute. Therefore, we recommend for other people planning such an event to anticipate large dropout numbers (up to 50%) and over-recruit accordingly. We found that the small group discussions were essential to ensure each participant had an opportunity to discuss their priorities and the use of facilitators worked well to ensure everyone had an equal voice.

Conclusions

Eleven different research priorities were generated through stakeholder-centred consensus. The 11 research questions encompass a range of categories, including: understanding the existing care patients receive; exploring what optimal care post-TIA/minor stroke should comprise and how that care should be delivered; impact on family members; and education/training for health care professionals. These research questions could usefully inform the research agenda for TIA/minor stroke. Inclusion of stakeholders in setting research priorities is important to increase the relevance of research and reduce research waste.

Abbreviations

GPs: General practitioners; QoL: Quality of life; TIA: Transient ischaemic attack; UK: United Kingdom

Acknowledgements

Thank you to all stakeholders and patients that participated in the priority setting consensus meeting.

Funding

The priority setting consensus meeting was funded by the Wellcome Trust Institute Strategic Support Fund (ISSF).

Availability of data and materials

The transcripts of the meeting are available from the corresponding author on reasonable request.

Authors' contributions

GMT, JM, TM and MC contributed to the study conception and design. GMT recruited the stakeholders and organised the meeting. GMT, RB and CM facilitated the meeting. GMT drafted the manuscript and RB, CM, JM, TM and MC provided feedback. All authors read and approved the final manuscript.

Authors' information

Not applicable.

Ethics approval and consent to participate

Ethical consent was gained from the University of Birmingham's Science, Technology, Engineering and Mathematics Ethical Review Committee (reference: ERN_16-1057). Written consent was obtained from all stakeholders prior to the meeting.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Received: 13 October 2017 Accepted: 10 January 2018 Published online: 25 January 2018

References

- Easton JD, Saver JL, Albers GW, Alberts MJ, Chaturvedi S, Feldmann E, Hatsukami TS, Higashida RT, Johnston SC, Kidwell CS, et al. Definition and evaluation of transient ischemic attack: a scientific statement for healthcare professionals from the American Heart Association/American Stroke Association stroke council; council on cardiovascular surgery and anesthesia; council on cardiovascular radiology and intervention; council on cardiovascular nursing; and the interdisciplinary council on peripheral vascular disease. The American Academy of Neurology affirms the value of this statement as an educational tool for neurologists. Stroke. 2009;40(6):2276–93.
- Fischer U, Baumgartner A, Arnold M, Nedeltchev K, Gralla J, De Marchis GM, Kappeler L, Mono ML, Brekenfeld C, Schroth G, et al. What is a minor stroke? Stroke. 2010;41(4):661–6.
- Scarborough P, Peto V, Bhatnagar P, Kaur A, Leal J, Luengo-Fernandez R, Gray A, Rayner M, Allender S. Stroke statistics. London: British Heart Foundation and Stroke Association; 2009.
- Rothwell PM, Coull AJ, Giles MF, Howard SC, Silver LE, Bull LM, Gutnikov SA, Edwards P, Mant D, Sackley CM, et al. Change in stroke incidence, mortality, case-fatality, severity, and risk factors in Oxfordshire, UK from 1981 to 2004 (Oxford vascular study). Lancet (London, England). 2004;363(9425):1925–33.
- Moran GM, Fletcher B, Feltham MG, Calvert M, Sackley C, Marshall T. Fatigue, psychological and cognitive impairment following transient ischaemic attack and minor stroke: a systematic review. Eur J Neurol. 2014;21(10):1258–67.
- Not just a funny turn: the real impact of TIA. https://www.stroke.org.uk/ sites/default/files/sa-the_real_impact_of_tia-report-sp-v5_0.pdf. Accessed 17 Jan 2018.
- Turner GM, Calvert M, Feltham MG, Ryan R, Marshall T. Ongoing impairments following transient ischaemic attack: retrospective cohort study. Eur J Neurol. 2016;23(11):1642–50.
- van Rooij FG, Kessels RP, Richard E, De Leeuw FE, van Dijk EJ. Cognitive impairment in transient ischemic attack patients: a systematic review. Cerebrovasc Dis (Basel, Switzerland). 2016;42(1-2):1–9.

- Carlsson GE, Moller A, Blomstrand C. Managing an everyday life of uncertainty–a qualitative study of coping in persons with mild stroke. Disabil Rehabil. 2009;31(10):773–82.
- Fride Y, Adamit T, Maeir A, Ben Assayag E, Bornstein NM, Korczyn AD, Katz N. What are the correlates of cognition and participation to return to work after first ever mild stroke? Top Stroke Rehabil. 2015;22(5):317–25.
- 11. Green TL, King KM. Functional and psychosocial outcomes 1 year after mild stroke. J Stroke Cerebrovasc Dis. 2010;19(1):10–6.
- Muus I, Petzold M, Ringsberg KC. Health-related quality of life among Danish patients 3 and 12 months after TIA or mild stroke. Scand J Caring Sci. 2010;24(2):211–8.
- Spurgeon L, James G, Sackley C. Subjective experiences of transient ischaemic attack: a repertory grid approach. Disabil Rehabil. 2013;35(26): 2205–12.
- Verbraak ME, Hoeksma AF, Lindeboom R, Kwa VI. Subtle problems in activities of daily living after a transient ischemic attack or an apparently fully recovered non-disabling stroke. J Stroke Cerebrovasc Dis. 2012;21(2):124–30.
- Carlsson GE, Moller A, Blomstrand C. Consequences of mild stroke in persons <75 years – a 1-year follow-up. Cerebrovasc Dis (Basel, Switzerland). 2003;16(4):383–8.
- Edwards DF, Hahn M, Baum C, Dromerick AW. The impact of mild stroke on meaningful activity and life satisfaction. J Stroke Cerebrovasc Dis. 2006;15(4):151–7.
- Radman N, Staub F, Aboulafia-Brakha T, Berney A, Bogousslavsky J, Annoni JM. Poststroke fatigue following minor infarcts: a prospective study. Neurology. 2012;79(14):1422–7.
- Hommel M, Trabucco-Miguel S, Joray S, Naegele B, Gonnet N, Jaillard A. Social dysfunctioning after mild to moderate first-ever stroke at vocational age. J Neurol Neurosurg Psychiatry. 2009;80(4):371–5.
- Wang YL, Pan YS, Zhao XQ, Wang D, Johnston SC, Liu LP, Meng X, Wang AX, Wang CX, Wang YJ. Recurrent stroke was associated with poor quality of life in patients with transient ischemic attack or minor stroke: finding from the CHANCE trial. CNS Neurosci Ther. 2014;20(12):1029–35.
- Faulkner J, McGonigal G, Woolley B, Stoner L, Wong L, Lambrick D. A randomized controlled trial to assess the psychosocial effects of early exercise engagement in patients diagnosed with transient ischaemic attack and mild, non-disabling stroke. Clin Rehabil. 2015; 29(8):783–94.
- Ihle-Hansen H, Thommessen B, Fagerland MW, Oksengard AR, Wyller TB, Engedal K, Fure B. Effect on anxiety and depression of a multifactorial risk factor intervention program after stroke and TIA: a randomized controlled trial. Aging Ment Health. 2014;18(5):540–6.
- Kamm CP, Schmid JP, Muri RM, Mattle HP, Eser P, Saner H. Interdisciplinary cardiovascular and neurologic outpatient rehabilitation in patients surviving transient ischemic attack or stroke with minor or no residual deficits. Arch Phys Med Rehabil. 2014;95(4):656–62.
- Lennon O, Blake C. Cardiac rehabilitation adapted to transient ischaemic attack and stroke (CRAFTS): a randomised controlled trial. BMC Neurol. 2009; 9:9.
- MacKay-Lyons M, Gubitz G, Giacomantonio N, Wightman H, Marsters D, Thompson K, Blanchard C, Eskes G, Thornton M. Program of rehabilitative exercise and education to avert vascular events after non-disabling stroke or transient ischemic attack (PREVENT trial): a multi-centred, randomised controlled trial. BMC Neurol. 2010;10:122.
- Prior PL, Hachinski V, Unsworth K, Chan R, Mytka S, O'Callaghan C, Suskin N. Comprehensive cardiac rehabilitation for secondary prevention after transient ischemic attack or mild stroke: I: feasibility and risk factors. Stroke. 2011;42(11):3207–13.
- Optimising Psychoeducation for Transient Ischaemic attack and MInor Stroke Management (OPTIMISM). https://www.nottingham.ac.uk/research/ groups/strokerehabilitation/projects/optimism.aspx. Accessed 17 Jan 2018.
- Boss HM, Van Schaik SM, Deijle IA, de Melker EC, van den Berg BTJ, Scherder EJA, Bosboom WMJ, Weinstein HC, Van den Berg-Vos RM. A randomised controlled trial of aerobic exercise after transient ischaemic attack or minor stroke to prevent cognitive decline: the MoveIT study protocol. BMJ Open. 2014;4(12):e007065.
- Rochette A, Korner-Bitensky N, Bishop D, Teasell R, White CL, Bravo G, Cote R, Green T, Lebrun LH, Lanthier S, et al. The YOU CALL-WE CALL randomized clinical trial: impact of a multimodal support intervention after a mild stroke. Circ. Cardiovasc Qual Outcomes. 2013;6(6):674–9.

- Berge E, Salman RA-S, van der Worp HB, Stapf C, Sandercock P, Sprigg N, Macleod MR, Kelly PJ, Nederkoorn PJ, Ford GA, et al. Increasing value and reducing waste in stroke research. Lancet Neurol. 2017;16(5):399–408.
- Chalmers I, Bracken MB, Djulbegovic B, Garattini S, Grant J, Gülmezoglu AM, Howells DW, Ioannidis JPA, Oliver S. How to increase value and reduce waste when research priorities are set. Lancet. 2014;383(9912):156–65.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

